

## **Author's response to reviews**

**Title:** Using Bayesian statistics to estimate the likelihood a new trial will demonstrate the efficacy of a new treatment

### **Authors:**

David Biau (david.biau@aphp.fr)

Samuel Boulezaz (samuel.boulezaz@gmail.com)

Laurent Casabianca (laurent.casabianca@aphp.fr)

Moussa Hamadouche (moussa.hamadouche@aphp.fr)

Philippe Anract (philippe.anract@aphp.fr)

Sylvie Chevret (sylvie.chevret@aphp.fr)

**Version: 2 Date:** 10 Jul 2017

### **Author's response to reviews:**

"The authors' response letter has been included as a supplementary file"